

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Sadig, Husam

Working Paper Unknown eligibility whilst weighting for non-response: The puzzle of who has died and who is still alive?

ISER Working Paper Series, No. 2014-35

Provided in Cooperation with: Institute for Social and Economic Research (ISER), University of Essex

Suggested Citation: Sadig, Husam (2014) : Unknown eligibility whilst weighting for non-response: The puzzle of who has died and who is still alive?, ISER Working Paper Series, No. 2014-35, University of Essex, Institute for Social and Economic Research (ISER), Colchester

This Version is available at: https://hdl.handle.net/10419/126481

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Unknown Eligibility whilst Weighting for Non-response: The Puzzle of who has Died and who is still Alive?

Husam Sadig Institute for Social and Economic Research University of Essex

No. 2014-35 October 2014



INSTITUTE FOR SOCIAL & ECONOMIC RESEARCH



Non-technical summary

Most panel surveys are designed to study a living population in a country. Sample members of panel studies are interviewed repeatedly over time. The conclusions arrived at through the sample are generalised to the whole population. Thus, such samples should be representative of the relevant populations. Otherwise, the results generated from the sample cannot be said to be correct with respect to the whole population.

Survey organisations usually prepare numerical quantities to be included in the data to compensate for sample members who do not participate. These quantities, called 'weights', increase the influence of participants who appear to be similar to those who did not participate. It is therefore important to be able to distinguish between non-participants and sample members who have died and are therefore no longer eligible to participate. One particular challenge that faces survey researchers in panel studies is that when a sample member stops participating at some point during the course of the survey it is not easy to be sure whether or not that person is still alive at any particular point in the future. Some of these non-participants might be dead in which case they must be excluded from the calculation of the weights. Otherwise, the characteristics of dead persons will mistakenly be used to increase the influence of those who participate and have similar characteristics. Hence the survey results may become misleading.

This research evaluates a method of estimating percentages of those who died in the sample amongst those who cease participation. The estimation is based upon using information on death percentages from the population. The estimated percentages are used to reduce the effect of those who died in the context of compensating for those who are still alive and did not participate. I used the British Household Panel Survey data, and I find that some of those who stop participating are dead. Most of the dead persons amongst those who stop participating are in the oldest age group in the sample. When this information is used in the compensation for non-participation, some results change considerably.

Unknown Eligibility whilst Weighting for Non-response: The Puzzle of who has Died and who is still Alive?

Husam Sadig

Institute for Social and Economic Research, University of Essex

Abstract

A challenge for longitudinal surveys is to update eligibility status over time for sample units, including those who cease participating. Ineligible units should be excluded from the base for weighting, as weighting would otherwise increase the influence of responding sample units that are similar to the ineligible ones. This paper estimates likelihood of eligibility of the British Household Panel Survey (BHPS) sample members by gender and age using population mortality data. This likelihood is then used to adjust non-response weights. Analysis is conducted using adjusted and unadjusted weights. Results show that the adjustment affects the significance level of some variables.

Keywords

Survey eligibility; population of interest; representative sample; non-response; weighting.

Author correspondence address: <u>hesadi@essex.ac.uk; husam.sadig@hotmail.com</u>.

Acknowledgements: Thanks are due to Peter Lynn and Olena Kaminska for their constructive comments and helpful advice.

1. Introduction

Longitudinal surveys follow respondents over time and continue gathering data from them, which not only enables the collection of up-to-date answers to the survey questions, but also allows the possibility of adding new questions. These complicated features of longitudinal surveys leads to a thorough understanding of dynamic populations, which cannot be achieved by conducting cross-sectional surveys (Lynn, 2009). However, to achieve this reward, a well designed representative sample is required. The sample selected for a longitudinal survey is usually designed to represent the population of interest at the start of the survey (Lynn, 2011). Surely, the population of interest changes overtime as people are born, immigrate, die and/or emigrate (Lynn, 2011). Thus, the sample is modified to maintain representativeness of the population of interest during the course of the study. New eligible members join the sample through a specific mechanism that is often established by the sample design¹. Accordingly, the survey researcher has control over the system by which new members join the sample, and hence new eligible sample members can be known. In turn, some sample members may die or move out of the scope of the survey, in which case they become ineligible and are no longer part of the population of interest. These ineligible participants should be identified and excluded from any analysis that aims to provide estimates of the population of interest. Otherwise, the sample used for estimation will not represent the study population correctly and will, consequently, provide biased estimates.

However, identifying eligibility status of survey participants over time can be a challenging task. For example, during the course of the survey, the survey organisation may lose track of some of the respondents, because, for instance, they have moved house without informing the survey conductors. Also, sometimes, even if the interviewer is successful in contacting a household, respondents may not want to take part in the survey. All of these cases, and any others where an interview cannot be conducted, result in non-

¹ The process by which new eligible members join the sample varies across surveys. For example, in Understanding Society in the UK, new born children of any eligible female sample member are added to the sample as eligible sample members (Lynn, 2011).

response (Sadig, 2011). Non-response obstructs the identification of eligibility status of survey participants, since little information about them is available. This is particularly a dilemma at waves subsequent to the initial non-response incident as information about non-respondents may not be available at all. For example, consider a survey that does not attempt to re contact sample members who refuse to participate in the survey when they are first contacted. At the first contact attempt (say the first wave), the survey researcher maybe able to identify which cases amongst those who refused to participate are actually eligible since the interviewer managed to contact them. However, it is almost impossible to classify the same cases as either eligible or ineligible at the, say, fifth wave when the last contact was made with them four waves ago.

As a result, eligibility status for a considerable number of non-respondents will be unknown. Therefore, estimating the eligibility status for respondents whose eligibility is unknown has puzzled survey researchers for the past few decades.

Unknown eligibility raises a number of practical concerns in longitudinal surveys. The major concerns are:

- (a) It disturbs the calculation of the survey quality measures such as response rate, contact rate and co-operation rate.
- (b) More importantly, it may spoil the reward of any weighting process endeavours to correct for non-response.

The existing methods (described in section 6) of estimating the proportion of eligible respondents among respondents of unknown eligibility are inadequate (Smith, 2003). Although these methods can be used to estimate the proportion of eligible respondents, which may help in calculating more accurate quality measures, they fail to assist in accurately calculating the weights. This is because estimating eligibility status is not done at the case level, and as a result, some ineligibles may be included in the creation of the weights.

This paper investigates an alternative approach to estimate eligibility status using population information from an external source. This approach allows for estimating proportions of eligible respondents by gender and age. The point of interest is to compare the estimated proportions of eligibility in the population with those in the sample so as to work out an adjustment coefficient that represents eligibility likelihood for sample members of uncertain eligibility for each category of gender and age group. This coefficient can then be used to modify non-response weights before assigning them to cases in the sample. As a result, the effect of weighting will be controlled by likelihood of eligibility of the respondent. This will reduce any bias that might potentially occur due to including ineligible units in the calculation of the weights.

Aside from survey-specific characteristics of ineligible respondents, the most common characteristics of being ineligible for a survey are: death, moving out of the geographical area that is covered by the survey and being institutionalised –such as going to prison-where contact cannot be made. In all these cases, conducting an interview is unfeasible. However, in this paper, the discussion is limited to death. Death is a special case of ineligibility since it is an absorbing state (i.e. once a respondent become ineligible through death, they can never be eligible again). Thus, the terms 'eligible' and 'ineligible' are used in this paper to refer to survival and death respectively.

The paper uses data from The British Household Panel Survey (BHPS) and mortality rates from the Office for National Statistics (ONS) to estimate eligibility rates.

The paper reviews the concept of eligibility and the effect of uncertain eligibility on the calculation of survey quality measures, but it mainly focuses on dealing with unknown eligibility in the weighting context.

2. Eligibility

The objective of sample surveys is to make inference about a population based on information obtained from the sample. Usually, the population of interest is defined precisely according to specific characteristics. Sample units whose characteristics match the characteristics of the population of interest are referred to as eligible sample units. Defining eligibility is a crucial step in every survey. Conditions for being eligible vary between surveys, depending on the aim and objectives of the survey. In many surveys, the definition of eligibility is linked to a certain period or point in time. For example, in a survey of smokers, if being eligible is defined as being a smoker, the survey organisation

should link the definition of eligibility to a specific time period, as individuals may start or stop smoking during the data collection period.

Since the only usable data for analysis are collected from eligible sample units, ineligible cases are dropped from the sample, and as a result, the sample size is then reduced. Thus, especially in cross-sectional surveys, it is advantageous to increase the eligibility rate, by, for example, pre-screening the sample units before selecting the sample. This is because, at the sampling stage, sometimes it is difficult for the survey researcher to spot some of the undesirable or ineligible cases (cases that are not part of the population of interest) in the sample frame. For instance, in random dialling digit surveys the sample frame may contain non-working numbers; however, it might be impossible to know this unless a contact attempt is made (Groves *et al*, 2004).

In contrast, in longitudinal surveys, pre-screening the sample may not be of great benefit in the long term. This is because individuals' characteristics that match the characteristics of the population of interest can change overtime, allowing respondents to be part of the population in the earlier waves but not in the later waves. For example, if being eligible in a survey is defined by living in the country where the survey is conducted, some participants may leave the country after participating in a number of waves, and, as a result of this, they become out of the scope of the population of interest. This complexity demonstrates that dealing with eligibility in longitudinal surveys is more problematic.

Although it is cost effective if ineligible participants are identified before they are issued an interviewer, usually ineligible units cannot be identified until the data collection is completed and the interview is conducted.

3. Unknown eligibility

The term 'unknown eligibility' is used to refer to the status where there is not sufficient information about a respondent to allow them to be identified as either an eligible or ineligible participant after the data collection stage is completed.

The most common outcomes of any contact attempt are completed interview, refusal or non-contact. In a successful interview, eligibility is usually known, since the interviewer is able to receive responses to at least a large part of the questionnaire. However, with non-response, which occurs through refusal or non-contact, information about nonrespondents is very limited and sometimes not available. Therefore, the survey researcher may be able to identify eligibility for some of the non-respondents, but for a substantial proportion, eligibility will remain unknown.

Unknown eligibility can be resolved in the case of wave non-response, where respondents are not present for at least one wave, but they resume participation at some point during the course of the survey. In this case, information related to eligibility status during the period of absence can be collected in the current interview. In turn, in panel studies, a special case of unknown eligibility occurs through attrition. Attrition is identified as the permanent dropout from a longitudinal survey after having participated at previous points of data collection (Chang, 2010). In this case, the survey researcher is unable to identify the eligibility status of sample persons even though they were eligible when they gave their last interview. Thus, it is desirable to reduce attrition by keeping track of respondents between waves (McGonagle *et al*, 2011; Laurie *et al*, 1999).

Nonetheless, despite the use of different strategies to minimize attrition in many studies (Laurie and Lynn, 2009; Laurie *et al*, 1999), in some cases it is impossible to retain survey participation. Death is one example of this and may be of particular concern in longitudinal surveys. If death is not reported to the survey organisation, dead respondents will be classified as respondents whose eligibility is unknown.

4. Unknown eligibility and response rate, contact rate and co-operation rate

When the data collection stage is completed, survey organizations usually publish some statistics such as the response rate, contact rate and co-operation rate, to reflect the main features of the data and inform data users about the quality of the data that the survey has gathered. However, each of these rates is defined as a ratio that contains the proportion of eligible sampled units in the denominator. Thus, an incorrect estimate of the proportion of eligible units amongst units of unknown eligibility will result in under or over estimating response, contact and co-operation rates.

4.1 Response rate

The response rate measures the percentage of the completed interviews² out of all the eligible units. For many surveys, response rates are calculated to examine the quality of the survey and the effort put forward to achieve the interviews. Additionally, the response rate may draw attention to investigating potential bias in the estimates. Usually, higher response rates are preferred, as they might reduce the potential bias introduced by non-response. However, the response rate on its own does not provide information about non-response error, but calculating the rate is a crucial stage in examining the presence of non-response error. The definition of the response rate implies that ineligible sample units should not be included in the calculation if the rate is to be computed accurately.

According to the Survey Research Centre (SRC), University of Waterloo (2005), the response rate (RR) is defined as

$$RR = \frac{\text{Number of eligible sample units with completed interviews}}{\text{Number of eligible sample units}}$$
(1)

However, it is almost impossible to calculate the denominator precisely, since the number of eligible cases among non-respondents cannot be identified exactly. In almost every survey, as long as there is an incidence of non-response, there will be a number of cases whose eligibility remains unknown. Therefore, the total number of sample units (TNSU) can be broken down into two components

TNSU= Number of units with known eligibility + Number of units with unknown eligibility

Consequently, in order to calculate the response rate, the survey researcher has to estimate the number of eligible units among the units of unknown eligibility (AAPOR, 2011). The number of estimated eligible sample units (NEESU) is a sub-group of the number of units

² The literature on response rate usually distinguishes between fully completed questionnaires and partially completed questionnaires. For the purpose of this research both types are considered as one category.

with unknown eligibility. Accordingly, the definition of the response rate in equation (1) can be rewritten as

$$RR = \frac{\text{Number of eligible sample units with completed interviews}}{\text{Number of eligible units of known eligibility + NEESU}}$$
(2)

Overestimating NEESU leads to underestimating the response rate, while underestimating NEESU results in overestimating the response rate. Therefore, regardless of the method used to estimate NEESU, it is advisable to utilize a value of NEESU that does not inflate the response rate and hence give a false sense of valuing the quality of the data.

4.2 Contact rate

The contact rate indicates the proportion of persons who were contacted by the interviewer, even if they refused to participate in the survey or were unable to provide any type of information (Gasteiz, 2007; SRC, 2005).

The contact rate (CR) is defined as

$$CR = \frac{\text{Number of eligible sample units in which contact was made}}{\text{Number of eligible sample units}}$$
(3)

Similar to the response rate, the denominator cannot be known precisely with the presence of non-response. Thus, to calculate CR, it is vital to estimate the number of eligible units amongst those of unknown eligibility. Yet again, this can be represented by NEESU. Thus, equation 3 can be rewritten as follows:

$$CR = \frac{\text{Number of eligible sample units in which contact was made}}{\text{Number of eligible units of known eligibility + NEESU}}$$
(4)

4.3 Co-operation rate

The co-operation rate measures the proportion of achieved interviews among the cases in which contact was made (Gasteiz, 2007; SRC, 2005).

The co-operation rate (CoR) is defined as

$$CoR = \frac{\text{Number of sample units in which interview was conducted}}{\text{Number of eligible sample units in which contact was made}}$$
(5)

In many social science surveys (such as the BHPS), a sample member is eligible if they are alive and living in the geographical area covered by the survey. In this case, the survey researcher does not need to calculate NEESU in order to calculate the co-operation rate. This is because the denominator in the co-operation rate only consists of sample members who are successfully contacted and hence eligible (i.e. alive and living in the geographical area covered by the survey).

However, using the contact rate and the co-operation rate, the response rate can be redefined as

Nevertheless, calculating the response and/or contact rates precisely requires the availability of the number of eligible sample units among non-respondents (NESU). Since it is not possible to count the NESU, it can be replaced with the number of estimated eligible sample units (NEESU). NEESU can be estimated using a number of practical methods. These methods are reviewed in section (6).

Thus, regarding the calculation of the survey quality measures, unknown eligibility can – to an extent- disturb the calculation of some of these measures. However, with a good estimation of NEESU, one can still calculate the response rate and the contact rate to the best possible approximation.

5. Unknown eligibility and weighting

In longitudinal surveys, apart from calculating precise quality measures, identifying the eligibility status of respondents will benefit the weighting process. Weighting is a process

by which a higher value is assigned to some of the eligible respondents in the survey, in order to modify them to represent eligible individuals who are missing due to non-response or an incomplete frame (Sadig, 2011; Biemer and Christ, 2008; Lynn, 2005). Thus, in order for the weights to modify the sample correctly, they should be calculated using eligible sample members only. Including ineligible sample members in the non-response model that is used to calculate the weights will lead to inaccurate sizes of the weights. For example, for a given survey where eligibility is defined as being alive, suppose that

 n_{ijk} indicates the number of sample units (where *i*, *j* and *k* denote eligibility status, knowledge of eligibility, and survey response status respectively; and that

i = 1 if eligible; 2 if ineligible (actual status, regardless of whether this is known);

j = 1 if eligibility status is known; 2 if it is not known;

k = 1 if survey respondent; 2 if non-respondent.

Thus,

$$\sum_{i=1}^{2} \sum_{j=1}^{2} \sum_{k=1}^{2} n_{ijk} = n_{\bullet\bullet\bullet} \text{ is the total sample size.}$$

We can assume that $n_{121} = n_{211} = n_{221} = 0$ (i.e. that all respondents are eligible and known to be eligible)

Thus,

 n_{111} is the number of respondents;

 n_{112} is the number of non-respondents known to be eligible;

 n_{212} is the number of non-respondents known to be ineligible; and

 $n_{122} + n_{222}$ is the number of non-respondents of unknown eligibility

Now, for a given weighting class 'c', the response probability (P_c) is calculated by dividing the number of responding units by the total number of respondents in the class. Accordingly, the relevant non-response adjustment weight (w_c) is calculated by the inverse of the response probability in the class.

Thus, in the presence of unknown eligibility:

$$P_{1c} = \frac{n_{111}}{n_{111} + n_{222}}$$
 leading to $W_{1c} = \frac{n_{111} + n_{222}}{n_{111}}$

But, with perfect information about eligibility, this should be:

$$P_{2c} = \frac{n_{111}}{n_{11.}}$$
 leading to $W_{2c} = \frac{n_{11.}}{n_{111}}$

Noticeably, $w_{1c} > w_{2c}$. Thus, with w_{1c} , cases in class 'c' will be over-weighted (i.e. if ineligible cases amongst cases of unknown eligibility are not identified as such and are excluded from weights' calculation). Moreover, the size of w_{1c} incorrectly increases as more cases are added to n_{222} . In other words, the relevant weight in a given class will mistakenly be increased, if more non-respondents ineligible cases are not identified in that class.

Accordingly, this is particularly a problem if ineligible cases among cases of unknown eligibility are not evenly distributed across weighting classes. In this case, respondents in weighting classes with larger proportions of unidentified ineligible cases, will have larger weights. As a result, cases in these weighting classes will be over-weighted. Consequently, weighted estimates will be biased towards characteristics from classes where more ineligible cases are not identified.

This is because the weights will increase the influence of responding units in these classes and hence will mistakenly boost the sample, by representing a proportion of individuals who are not part of the population of interest. Consequently, estimates resulting from such a weighting strategy are biased. With regard to ineligibility through death, health studies have shown that death is associated with socio-demographic characteristics such as age and gender (Singh-Manoux *et al*, 2008; Dr Foster, 2004). That is to say, in most parts of the world women are expected to live longer than men, and mortality rates are higher among older age groups than among their younger counterparts. Therefore, death does not happen completely at random. Hence, results based on a weighting strategy that mistakenly include dead respondents in the weights' calculation may be biased towards the categories of respondents that have higher death rates.

Nonetheless, it is not always possible to be aware of the death of non-respondents, especially if the respondent is a single-person household. In this case, the death of the respondent may not be reported since there is no other household member to do so.

There is an increasing interest in the issue of weighting under uncertain eligibility. However, at present, we do not know of any attempt that deals with uncertain eligibility while weighting to correct for non-response. In most longitudinal surveys such as the British Household Panel Survey (BHPS), Swiss Household Panel (SHP) and the German Socio Economic Panel (GSOEP) weighting assumes that individuals whose eligibility is uncertain are eligible (Taylor *et al*, 2010; Plaza and Graf, 2008; Kroh, 2009).

However, eligible respondents can be identified in proportions by one or two categorical variables (e.g. age group and ethnicity). These proportions across the two selected variables can be estimated using information from population data. Such information could be available from an external source (e.g. census). The same proportions can be calculated in the sample. Comparing these proportions of eligibility in the sample with those in the population can disclose the degree at which eligibility rates are miscalculated in the sample. Furthermore, this comparison, since it is done across the categories of two variables can assist in identifying the likelihood that a respondent in a given category is eligible. As a result, non-response weights in a given category can be controlled by the likelihood of eligibility for respondents in that category. This strategy of weighting (explained in section 7) is the subject of this paper.

6. Methods of estimating the proportion of eligible cases

There are several methods which are usually used to estimate the rate of cases of unknown eligibility that are actually eligible 'e'. Most of the literature in this area assumes a random digit dialling survey (RDD) (Smith, 2003). Therefore, some of the methods are RDD specific.

Minimum and maximum allocation: in this method 'e' is assumed to be either 0% or 100% of the cases of unknown eligibility (Lessler and Kalsbeck, 1992; smith 2003). Accordingly, more than one response rate can be produced. Taking 'e' as 0%, gives the maximum possible response rate while substituting 'e' as 100% produces the lowest response rate. Smith (2003) indicates that this method is only useful in determining the upper and lower bounds of the response rate. However, one can obtain a range of rates by varying the values of 'e' from 0% to 100% before choosing a plausible value that does not inflate the response rate. However, if the level of unknown eligibility is high, the number of the possible response rates will become impractical.

*Proportional allocation*³: this method assumes that 'e', among the cases of unknown eligibility, is the same as among the cases whose eligibility is known (Frankel, 1983; Lessler and Kalsbeck, 1992; Smith, 2003; Barron, Khare and Zhao, 2008). Smith (2003) states that proportional allocation is conservative as it produces a high value of 'e', and hence does not inflate the response rate. However, he argues that it might produce a biased estimate of 'e' because it assumes that the eligibility rate among the unobserved sample is the same as among the observed sample.

Survival analysis: this method is the standard survival analysis method in which the number of contact attempts is used to estimate the eligible cases among the cases of unknown eligibility (Frankel *et al*, 2003; Smith, 2003). This method is considered to be a better approach to estimating 'e', since it uses more information from the sample than the other methods. However, Smith (2003) argues that one cannot be certain that the statistical assumptions of survival analysis are properly met.

RDD specific methods: there are a few methods used in random digit dialling surveys to estimate the eligibility rate among the unknown eligibility cases. The most commonly used of these are: allocation based on disposition codes and contacting telephone business offices. Under the disposition codes allocation approach, the outcome of the call attempt

³ Some of the literature on the response rate refers to this method as CASRO type II as it is proposed by the American Survey Research Organisations.

is used to identify whether a case is eligible or not (Smith, 2003). For example, in a survey, a researcher might establish a rule that all of the phone numbers with answering machines are eligible, while those resulting in busy signals are not eligible. The limitation of this method is that the basis in which the disposition codes are allocated may not solely determine the eligibility. For example, a ring-no-answer alone is not enough to identify a case as being ineligible.

As for the business offices approach, survey researchers sometimes contact local telephone business offices to enquire about the status of the unknown numbers (Smith, 2003; Frankel *et al*; 2003). However, this method is considered to be both money and time consuming, in addition to the fact that business offices usually refuse to give out information about phone numbers.

Many studies have applied the above methods to estimate the eligibility rate among the cases of unknown eligibility. For example, Barron, Khare and Zhao (2008) applied the proportional allocation approach to estimate 'e', to calculate the response rate for the National Immunization Survey's Cell Telephone Pilot study (NIS-CTP). Gasteiz (2007) indicates that the minimum and maximum allocation method was used ('e' was assumed to be 100%) to estimate the eligible cases among those cases where eligibility is unknown in the Population in Relation to Activity Survey (PRA). In a list-assisted RDD telephone survey about adolescent substance abuse, the Survey and Evaluation Research Laboratory (SERL) applied the proportional allocation approach to estimate the response rate (Ellis, 2000).

However, each of the methods used has its limitations, and as Smith (2003) states "At present none can be considered a gold standard for calculating "e"". Furthermore, there is no evidence to show that applying any of these methods identifies the eligible cases among cases of unknown eligibility in order to drop ineligible cases and calculate non-response weights appropriately. The focus has, instead, been on calculating survey response rates. Moreover, all of these methods have mainly been implemented in cross-sectional studies. In longitudinal surveys, besides the calculation of the eligibility proportion to compute response rates, the investigation of an alternative method that takes the longitudinal aspect of eligibility into account and utilizes information from inside and

outside the survey, could assist in identifying eligible cases, and may therefore result in successful weighting.

7. Methodology

The data used in this paper are from waves 1 to 18 of the BHPS, covering the years 1991 to 2008. The analysis was done at the individual level and was restricted to the original sample of BHPS and respondents aged 16 or older. The estimation was implemented in STATA.

The sample –if representative- is a smaller image of the population. Thus, rates of phenomena in the populations should be equal to those of the same phenomena in the sample, under expectation. One concern of this paper is to compare the proportion of eligibility in the sample with this in the population. This will be done for categories of respondents (by gender and age group). An eligibility rate for a given category in the sample will be compared with the eligibility rate in the equivalent category in the population.

Eligibility in the sample is either known or unknown. If both eligible respondents and those with unknown eligibility in the sample are considered as eligible respondents, the calculated eligibility rates in categories with larger proportions of unknown eligibility cases in the sample will be higher than the rates in the equivalent categories in the population (because some of the unknown eligibility cases in the sample may not be eligible). Accordingly, in such categories, non-response weights will not correct the sample proportions in a manner that make them represent their equivalent proportions in the population unless the weights are adjusted. Therefore, an adjustment factor is needed for the categories in the sample to modify the relevant weights so that weighted proportions in the sample represent the equivalent proportions in the population. This adjustment factor can be worked out based on the comparison between eligibility proportions in the sample and the population.

Letting P_P and P_S be the eligibility proportion in the population and eligibility proportion in the sample respectively. If unknown eligibility cases in the sample are considered eligible, then

$$\mathbf{b}_{S} \ge \mathbf{b}_{P} \tag{7}$$

if

$$ad * \mathfrak{P}_{S} = \mathfrak{P}_{P} \tag{8}$$

Then *ad* is a fraction.

ad is the adjustment factor that equalises the eligibility proportions in the sample and the population. From equation 8

$$ad = \mathbf{P}_P / \mathbf{P}_S \tag{9}$$

In other words, the adjustment factor is the ratio of the eligibility proportion in the population to the eligibility proportion in the sample. This ratio takes a value between 0 and 1 since its denominator is larger, and should be used to adjust non-response weights. The lesser differences between eligibility proportions in the sample and the population the closer the value of *ad* to 1 and hence the smaller the adjustment on weighting will be and vice versa. For example, in a given population, assuming that eligibility rate is 80% (i.e. 20% ineligible) and a representative sample from this population shows that 50% known eligible, 10% known ineligible and 40% unknown eligibility. If the weighting assumes that those with unknown eligibility are eligible, weights will increase the values of the responding sample to represent 90% eligibility in the population). Thus, the weights need to be adjusted so that they only represent the eligible respondents in the population. Based on equation 9, an adjustment of 8/9 is needed in this example. Section 11 explains in details how this adjustment factor was made for the BHPS sample.

While proportions of eligibility in the sample can be calculated from the data, proportions in the population should be estimated using information from external source. Note that eligibility in the BHPS is met if the sample member is alive and living in the United Kingdom (UK). Thus, eligibility proportions are actually the survival proportions. Therefore, this paper used information from The Office for National Statistics (ONS) on survival/mortality rates for the population in England and Wales. These were used to estimate proportions of eligibility in the population. However, the original sample of BHPS includes respondents from England, Wales and Scotland. Nonetheless, ONS does not publish annual mortality rates for the population in England, Wales and Scotland together (to match the target population of BHPS original sample). Thus, in this paper it was assumed that mortality rates for the population in England and Wales are the same as those for the population in England, Wales and Scotland. Based on this assumption, survival/mortality rates by ONS are considered to be for the target population of the BHPS original sample.

8. Calculating proportions of survivals in the sample

The BHPS data provide details about the contact outcome at every wave through a variable named *'individual interview outcome'* (IVFIO). The main categories of this variable indicate whether the outcome is full interview, proxy interview, telephone interview, refusal, in institution, non-contact, other non-interview, out of scope or dead. Thus, this setting leads to three categories of respondents in terms of eligibility status:

- (a) Eligible respondents (ER): these are respondents who gave a full interview, proxy interview, telephone interview, or refusal.
- (b) Ineligible respondents (IR): these are respondents who are reported dead, in institution or out of scope.
- (c) Respondents with unknown eligibility (RUE): these are mainly respondents who were not contacted.

Proportions of category (a) through to (c) were calculated by gender and age groups⁴ and are shown in table (1). For example, 44.36% of the males aged between 16 and 19 who joined BHPS in its first wave (1991) are still eligible in 2008 while only 1% of them are known to be ineligible and the remaining 54.64% are of unknown eligibility. Of particular interest, for both males and females, none of the respondents aged 85 or over at the start of the survey (1991) are known to be eligible in 2008 (0.00%). However, only 86.76% and 84.27% males and females (respectively) aged 85 or over are actually reported as being ineligible, and the rest are of unknown eligibility. Those who started the survey at

⁴ The age groups used were 16-19, 20-24, 25-34, 35-44, 45-54, 55-64, 65-74, 75-84, 85 and over. These are the same age groups used by the ONS to publish mortality rates for England and Wales. With the exception of the first age group as ONS provides mortality rates from the age group 15-19. However, the analysis here is restricted to those aged 16+.

the age of 85 or over and are of unknown eligibility 18 years later, are likely to be ineligible.

For all ages, amongst the males, 42.27% are eligible respondents, 15.79% are ineligible and 41.94% are of unknown eligibility. For the females, 48.28% are eligible respondents, 15.42% are ineligible and 36.3% are of unknown eligibility. This shows that the known eligibility rate for females is higher than for males, meaning that the unknown eligibility rate among females is lower than among males. The sample also shows interesting distributions of gender and age groups in terms of eligibility status. For instance, for both males and females, eligibility rate goes up as age increases, but reaches its peak in the age group 35-44 (51.03% M; 58.59% F) before it starts declining as age increases, to reach its nadir at the age of 85 or over (0.00% M; 0.00% F). This pattern of eligibility is reasonable, since mortality is higher amongst older age groups. Furthermore, ineligibility rate increases constantly with age. However, it increases faster for males than females and reaches its peak at age 85 or over (86.67% M; 84.27% F). This finding is consistent with the literature on mortality, which establishes that death rates are always higher amongst older age groups (Singh-Manoux et al, 2008) and life expectancy among females is higher than among males (Dr Foster, 2004). Additionally, for both males and females, the unknown eligibility rate decreases with increasing age, and its highest rate is in the age group 16-19 (54.64% M; 50.90% F).

	16-19	20-24	25-34	35-44	45-54	55-64	65-74	75-84	85+	All
										ages
Male										
Eligible	44.36%	41.74%	48.63%	51.03%	48.64%	43.23%	21.09%	3.21%	0.00%	42.27%
Ineligible	1.00%	1.06%	1.55%	4.42%	10.00%	26.10%	54.88%	79.12%	86.67%	15.79%
UE	54.64%	57.20%	49.82%	44.55%	41.36%	31%	24.03%	17.67%	13.33%	41.94%
<u>Female</u>										
Eligible	49.10%	57.62%	58.52%	58.59%	57.35%	49.23%	26.48%	4.75%	0.00%	48.28%
Ineligible	0.00%	0.21%	1.58%	2.55%	7.53%	19.60%	43.84%	69.13%	84.27%	15.42%
UE	50.90%	42.17%	39.90%	38.86%	35.12%	31.17%	29.68%	26.12%	15.73%	36.30%

Table 1 Proportions of eligible, ineligible and unknown eligibility cases in the original sample (1991) of BHPS in 2008 by gender and age groups

* UE refers to unknown eligibility.

However, whilst weighting, respondents of unknown eligibility (RUE) must be decided as either alive or dead (i.e. eligible or ineligible). For the purpose of the analysis in this paper, RUE were assumed to be eligible⁵. Therefore, eligible respondents (ER) and RUE were combined to calculate the proportions of survival in the sample. These proportions were calculated by age groups and gender and are shown with their corresponding proportions in the population in table (2).

9. Calculating proportions of survival from population information

ONS publishes annual mortality rates (ONS, 2014) by gender and 10-year age groups (with the exception of the first two age groups: 15-19 and 19-24) for the population in England and Wales (see these rates in appendix B and D). Using this information, the survival proportions by gender and age groups were calculated as follows:

• A rate in a given year for a certain category of age group and gender gives the probability that an individual in that category will die in the following year, given that he or she is alive in the current year. Subtracting the rates from 1 gave the survival probabilities in the following year (see these rates in appendix C and E). Rates for the years 1991 to 2007 were used to correspond to the 17 waves following the first wave (1992-2008). To simplify the calculations, the rates were expanded by a single year of age from age 16 to age 112⁶, by giving each age in the same age group the survival rate for that age group. Considering the original sample at wave 1 in 1991, the survival probabilities by the end of wave 18 (2008) were then calculated for each single year of age, by computing the product of the rates in the consecutive 17 years for each age. These probabilities were then regrouped into the original age groups by taking the average. These proportions are shown with their corresponding proportions from the sample in table (2).

⁵ In the standard approach of weighting, RUE are usually assumed to be eligible, especially if weighting is done through a model-based method.

⁶ The oldest participant in the original sample of BHPS was aged 96. By the end of the 18 waves, this person would have been aged 112 if they were still alive. Thus it was necessary to expand rates up to age 112, in order to be able to calculate the survival probabilities for all respondents.

Table 2 shows the proportions of survival in the sample and the population during the period 1991 to 2008 by gender and age group. For example, 2.73% of the males aged 85 or over in the population in 1991 are expected to still be alive by 2008 while the corresponding proportion of this in the sample is 13.33%. Overall, the survival proportions in the sample are larger than those in the population. This confirms the hypothesis that there is under reporting of death in the sample. Nevertheless, the differences between the survival proportions in the population and the sample are not worryingly large, and suggest only a small number of unreported deaths. The largest differences for both gender types are registered for the age groups 65-74, 75-84 and 85 or over. Interestingly, respondents in the oldest age group (85 or over) show smaller differences between the survival proportions in the population and the sample than the age groups 65-74 and 75-84. This may be because individuals aged 85 or over are less likely to be single-person households (at old age people usually need to be cared for either through family members or by professional carers) than those in the age groups 65-74 and 75-84 (at these ages individuals may still live independently as single-person households). Hence, death at age 85 or over has more chance of being reported than death in the two age groups, 65-74 and 75-84.

Table 2 Calculated survival proportions in 2008 in the population and assumed	proportions
in the original sample of BHPS by gender and age group	

	16-19	20-24	25-34	35-44	45-54	55-64	65-74	75-84	85+
Male									
Population	98.50%	98.16%	97.25%	93.69%	84.08%	61.39%	28.03%	6.07%	2.73%
Sample	99.00%	98.94%	98.45%	95.58%	90.00%	73.90%	45.12%	20.88%	13.33%
Female									
Population	99.36%	99.10%	98.35%	95.66%	89.74%	73.31%	40.58%	10.86%	5.43%
Sample	100.00%	99.79%	98.42%	97.45%	92.47%	80.40%	56.16%	30.87%	15.73%

*Entries are the survival proportions in 2008 for those who were alive in 1991. The sample proportions include both those who are known to be alive (eligible) and those with unknown status of survival (unknown eligibility).

Before describing how the adjustment factor was calculated for each category of gender and age group, the next section provides details of the creation of non-response weights.

10. Weights creation

Non-response weights were created based on respondents in the 18 waves using a modelbased method (the standard approach). The response propensity was modelled using logistic regression. The dependent variable was a categorical variable with two categories indicating whether a respondent participated in all of the 18 waves or not. Although the relationship between the response propensity and weighting variables may vary among waves, one way of creating non-response weights is to ignore the effect of time-varying variables (Sadig, 2011) and use variables from wave 1. This way guarantees availability of information for both respondents and non-respondents in the 18 waves, which is essential in modelling the response propensity. In this paper, a large combination of continuous and categorical variables from wave 1 was used to estimate the model. These variables⁷ were selected to represent three categories of variables: interview/interviewer condition/characteristics (e.g. interviewer's sex and length of interview), household characteristics (e.g. household size and household type) and individual characteristics (e.g. age, sex and savings). These variables are commonly used in the analysis of nonresponse (Urig, 2008; Nicoletti and Peracchi, 2005; Nicoletti and Buck, 2004).

$$\text{Logit}(R_i) = f\left(\sum_k I_k + \sum_{jk} H_{jk} + \sum_{ijk} D_{ijk} + \varepsilon_i\right)$$
(10)

Where:

 $R_i \equiv$ Responding Status at the 18 waves.

$$R_{i} = \begin{cases} 1, & \text{if unit i responded in all of the 18 waves} \\ 0, & \text{otherwise} \end{cases}$$
(11)

 $I_k \equiv$ Interview/Interviewer condition/characteristics.

 $H_{ik} \equiv$ Household characteristics.

 $D_{ijk} \equiv$ Individual characteristics.

 $\varepsilon_i \equiv$ Error term.

⁷ These variables are sex, age, ethnic group, region, health status, household size, presence of children in household, housing tenure, income, number of people age 75+ in the household, type of household, number in employment in household, education, employment status, savings, debt, type of accommodation, financial situation, socioeconomic group, number of weekly working hours, number of weekly overtime hours, work location, smoking status, car ownership, number of own children in the household, presence of others during interview, interviewer ID, interviewer sex and length of interview.

Table 3 shows the results of modelling the response propensity in the 18 waves through a logistic regression model. The table presents odds ratios. Regarding the tendency to respond in the period of 18 waves, most variables have a significant effect on response propensity. For example, females are more likely to respond than males ($\hat{b} = 1.393$, p < 0.001). In addition, respondents from a white ethnic origin tend to participate more than respondents from other ethnic origins ($\hat{b} = 1.652$, p < 0.01). Bad health however, appears to be negatively correlated with the response propensity ($\hat{b} = 0.839$, p < 0.05), indicating that individuals with better health have a greater tendency to respond. Also, homeowners are more likely to respond than non-homeowners ($\hat{b} = 1.91$, p < 0.05), while increase in age is negatively associated with survey participation ($\hat{b} = 0.994$, p < 0.05).

Additionally, the response propensity in the 18 waves is also significantly correlated (positively or negatively) with other variables. These are household type, education, employment status, savings ownership, housing tenure, sex of interviewer and region.

However, other factors, such as household size and the presence of children in a household do not show a significant effect on response propensity.

	Model of response propensity
	based on variables from wave1
Female	1.393***
White	1.652**
Bad health	0.839*
Household size	0.975
Household with dependent children	1.084
Home owner	1.191*
Age	0.994*
Annual income/1000	1.004*
Household has no member aged 75 or over	0.890*
1 or 2 persons in employment in household	0.855*
3 persons or more are in employment in	0.866
household	
Single person household	0.689**
Has GCE qualification or above	1.375***
Employed	1.367***
Having a second job	0.772*
Has no savings	0.818*
Living in a flat	0.635***
Based in business premises	0.948
Living in a bedsit	0.503
Living in other housing type	1.376
Interviewed by a female	1.462**
Lives in South-East	1.174
Lives in South-West	1.120
Lives in East Anglia	1.290
Lives in the Midlands	1.224
Lives in the North	1.325*
Lives in Wales	1.321
Lives in Scotland	0.902
N	10248
Pseudo R ²	0.087

Table 3 Logistic regression model of the response propensity in the 18 waves.

Note: The entries are odd ratios. The reference categories of the categorical independent variables in the model are male, non-white, good health, household with no children, not a home owner, there is at least one person aged 75+ in HH, no one is in employment in HH, multi-person household, does not have a GCE or higher degree, unemployed, has a second job, has savings, living in a house, interviewed by a male and lives in London respectively. * p < 0.05, ** p < 0.01, *** p < 0.001.

Accordingly, weights were calculated as the inverse of the predicted probabilities from the model for each case in the sample.

 $w_i = 1/r_i$

Where:

22

(12)

 $w_i \equiv \text{Case i initial non-response weight.}$

 $r_i \equiv$ Predicted probability of response for case i from the model.

The set of initial non-response weights was then multiplied by the BHPS set of design weights (w_D) . However, in the BHPS, the design weights were combined with a set of weights that compensate for wave 1 non-response. Thus, w_D , which is provided by BHPS, adjusts for the difference in the selection probabilities and compensate for wave 1 non-response simultaneously.

$$W_{NRi} = W_i^* W_{Di} \tag{13}$$

Where:

 $w_{NRi} \equiv \text{Case } i \text{ non-response weight.}$

 $w_i \equiv \text{Case } i \text{ initial non-response weight.}$

 $w_{Di} \equiv \text{Case } i \text{ design weight (combination of design weight and wave 1 non-response weight).}$

11. Adjustment factor

Theoretically, survival proportions in the sample should equal survival proportions in the population from which the sample is drawn if the sample is correctly representing the study population. However, since RUE in the sample were considered eligible (alive), proportions of death in the sample were under calculated. Consequently, this led to over calculating survival proportions in the sample. Thus, survival proportions calculated from the sample are higher than survival proportions estimated using the population information. These differences are due to not being able to identify some of the deaths in the sample. Based on the differences between survival proportions in the sample and the population, an adjustment factor that varies by gender and age groups was calculated (as shown in equations 7 to 9) as follows:

• For each age group and both genders, the adjustment factor was calculated as the ratio of proportion of survived respondents in the population to the proportion of survived respondents in the sample, under the assumption that all RUE are eligible. Since survival proportions in the sample are bigger than their correspondence proportions in the population, this ratio takes a value between zero and one. For instance, if, in any category of gender and age group in the sample, large amount of death is identified, survival proportions in the sample and the population will be approximately equal. Hence, the adjustment factor will take a value close to one and consequently lead to a small effect or no effect at all on the relevant weight. Contrary, if, large amount of death is in the population and accordingly the adjustment factor will take a value close to zero. Thus, the effect of the relevant weight will be reduced to represent lesser units from this category due to the low likelihood of eligibility. Hence, the adjustment factor represents the likelihood of survival for each category of gender and age group.

$$ad_{jk} = \mathbf{P}_{Pjk} / \mathbf{P}_{Sjk} \tag{14}$$

Where:

 $ad_{jk} \equiv$ Adjustment factor for the category of age group *j* and gender *k*.

 $P_{Pjk} \equiv$ Population proportion of survived respondents in age group *j* and gender *k* (probability that a respondent in age group *j* and with gender *k* is alive by the end of wave 18).

 $P_{Sik} \equiv$ Sample proportion of assumed survived respondents in age group *j* and gender *k*.

Thus, the final non-response weight for case i which falls in the category of the age group j and gender k, was calculated as the product of case i non-response weight and the adjustment factor in the category of age group j and gender k.

$$w_{FNRijk} = w_{NRijk}^* a d_{jk} \tag{15}$$

Where:

 $w_{FNRijk} \equiv$ Final non-response weight for case *i* in the category of age group *j* and gender *k*.

 $w_{NRijk} \equiv$ Non-response weight for case *i* in the category of age group *j* and gender *k*.

 $ad_{ik} \equiv \text{Adjustment factor for age group } j$ and gender k.

The adjustment factor should modify the weight of respondents in a given category according to the likelihood of survival by the end of the 18-wave period in that category. Accordingly, respondents in older age groups are more likely to receive smaller size weights than those in younger age groups, since older respondents have a lower chance of survival after 18 years. Thus, any potentially dead (ineligible) respondents who were initially included when weights were calculated will be removed from the adjusted weights depending on the survival likelihood in their category.

For example, table 2 shows that for women aged 85 or over, the survival proportions in the population and the sample are 5.43% and 15.73% respectively. Thus, the adjustment factor for respondents in this category can be calculated, based on equation 14, as follows:

 $ad_{85w} = 5.43/15.73$

 $ad_{85w} = 0.35$

This means that, while the sample shows that the survival proportion of women aged 85 or over at the beginning of BHPS (1991) in 2008 is 15.73%, the adjustment factor shows that only 35% of this proportion is likely to be alive in 2008. Accordingly, this will be used to adjust the weight for women in the responding sample who fall in this category. Therefore, the final non-response weight for women in this category can be calculated based on equation 15 As follows:

 $W_{FNR,85 w} = W_{NR,85w} * 0.35$

Similarly, for women in the age group 16-19, the table 2 indicates that their survival proportions in the population and in the sample are 99.36% and 100% respectively. Thus, the adjustment factor and final nonresponse weight for this category are

 $ad_{16-19w} = 99.36/100$

$$ad_{16-19w} = 0.99$$

$$W_{FNR,16-19 w} = W_{NR,16-19 w} * 0.99$$

Consequently, the weights for responding women in the age group 85 or over will be reduced more than the weights for responding women in the age group 16-19 since for the latter the extent to which the likelihood of survival is over estimated is smaller $(ad_{16-19w} = 0.99 > ad_{85w} = 0.35)$.

Table 4 shows all of the calculated adjustment factors. These are presented for each category of gender and age group. The table was calculated based on the information in table 2 and equation 14. As can be noticed from the table, for both men and women, the adjustment factor will have very small effect (mostly no effect) on weights for respondents aged between 16 and 64 ($0.83 \le ad \le 0.99$). However, for both men and women, the factor shows a drastic change for the ages above 64 implying more effect of the adjustment factor on the weights for respondents aged 65 or above ($0.20 \le ad \le 0.62$). Thus, adjusted weights are expected to have more impact on estimates related to older respondents (aged 65 or above) than on estimates based on their younger counterparts (aged between 16 and 64).

	16-19	20-24	25-34	35-44	45-54	55-64	65-74	75-84	85+
Male									
ad	0.99	0.99	0.99	0.98	0.93	0.83	0.62	0.29	0.20
Female									
ad	0.99	0.99	0.99	0.98	0.97	0.91	0.72	0.35	0.35

Table 4 Calculated adjustment factors for the categories of gender and age groups.

*Entries are the adjustment factors (ad). These are calculated based on equation 2.7 and information in table 2.2.

12. Assessment of the effect of the adjustment factor

Assessing the effect of the adjustment factor was done by conducting different types of statistical analyses using the two sets of weights (adjusted and unadjusted) and comparing the results. The analyses include calculating the coefficient of variation (CV) for adjusted and unadjusted weights as well as applying other data analysis techniques. The latter

includes producing descriptive statistics and estimating panel data models (multivariate analysis) using adjusted and unadjusted weights.

Since the effect of the adjustment factor is expected to be different for respondents aged 65 or older than those aged below 65, the analysis was done separately for respondents aged: between 16 and 64, 65 or older (65+) and for the whole sample.

12.1 Coefficient of variations (CV)

CV is a measure of dispersion. It is defined as a ratio of the standard deviation to the mean. Higher values of CV indicate larger variability in the data and vice versa. Since the calculation of CV does not directly involve the measurement unit (only uses the standard deviation and the mean), it can be used to compare the variability of two different variables. Thus, calculating and comparing CV for the adjusted and unadjusted weights will show whether the adjustment factor reduces the variance of the weights.

Table 5 shows the standard deviation, mean and CV for the unadjusted and adjusted weights. The results are presented separately for the full sample, respondents aged 16 to 64 and respondents aged 65+. As can be seen from the table, for both the whole sample and respondents aged 65+, the CVs for the unadjusted weights are higher than the CVs for adjusted weights ($CV_{un. \ sample} = 61\% > CV_{ad. \ sample} = 54\%$ and $CV_{un. \ 65+} = 71\% > CV_{ad. \ 65+} = 62\%$). These results indicate that, for the whole sample and respondents aged 65+, there is less variability in the sets of adjusted weights than in the unadjusted weights. In other words, the adjustment factor reduces the sizes of the largest weights amongst the weight values of respondents aged 65+ and the full sample. Based on this result, one may expect the standard errors of estimates to differ if adjusted weights are used in the estimation instead of unadjusted weights.

As for weights concerning respondents aged between 16 and 64, the CV for unadjusted weights ($CV_{un.\ 16-64}$ = 67%) is only 1% higher than the CV for adjusted weights ($CV_{ad.\ 16-64}$ = 66%).

	All responder	nts aged 16+	Respondents	aged 16 to 64	Respondents aged 65+		
SHS	Unadjusted weights	Adjusted weights	Unadjusted weights	Adjusted weights	Unadjusted weights	Adjusted weights	
C4J Door	1.20	1.02	1.21	1.07	2 60	1 1 2	
Sta.Dev	1.59	1.05	1.21	1.07	5.09	1.15	
Mean	2.28	1.92	1.81	1.61	5.19	1.81	
CV	0.61	0.54	0.67	0.66	0.71	0.62	

Table 5 Standard deviations, means and coefficients of variation for adjusted and unadjusted weights.

• Std.Dev is the standard deviation. CV is calculated as the ratio of Std.Dev to the Mean (CV=Std/Mean).

Turning to the descriptive and multivariate analyses, this was carried out to investigate the subjective health status (SHS) in the BHPS. The following two sections summarise these analyses.

12.2 Descriptive statistics

In the BHPS, SHS is measured by asking respondents every year to rank their own health as excellent, good, fair, poor or very poor. The proportions of respondents in each of these categories are calculated using adjusted and unadjusted weights and are displayed in table 6. The weighted proportions are presented for the whole sample, respondents aged 16 to 64 and respondents aged 65+ separately. Also, the table presents 95% Confidence Intervals (CI) for the unadjusted proportions. CI's are used here to assess whether the adjusted proportions are within the calculated CI's of the relevant unadjusted proportions. If any adjusted proportion falls out of the CI of its corresponding unadjusted proportion, this may be taken as an indication of a significant difference between the two proportions, and hence clear effect of the adjustment factor.

Focussing on respondents aged 16 to 64 first, adjusted and unadjusted weights produced similar proportions across the categories of SHS and none of the adjusted proportions seem to be out of the CI's of the unadjusted proportions. Thus, this result indicates no significant differences between adjusted and unadjusted proportions for those aged 16 to 64. In other words, there is no evidence of considerable differences between adjusted and unadjusted weights for the age group 16 to 64.

Turning to all respondents aged 16 and over, there is one significant difference relating to the category 'very poor health'. In this category, the adjusted proportion (1.12%) falls out

of the CI (1.55%-2.27%) of the unadjusted one (1.91%) indicating a significant difference due to making the adjustment.

As for those aged 65+, there are two significant differences here. First, using unadjusted weights, the proportion of those who reported poor health is (2.22%) with 95% CI of (1.45%-2.99%), meanwhile using the set of adjusted weights yields a lower percentage (1.40%) that is out of the range of the calculated CI. Second, proportions of those who reported very poor health are also significantly different. With adjusted and unadjusted weights, these proportions are 0.59% and 2.25% respectively. However, the adjusted proportion is lower than the lower limit of the CI of the unadjusted proportion (1.48%-3.02%).

Based on these results, it can generally be concluded that, for the whole sample and those aged 65+, adjusted weights produce different proportions than unadjusted weights; meanwhile, for those aged 16 to 64 adjusted and unadjusted weights result in similar proportions. This maybe explained by the fact that, when adjusting the weights, the effect of the adjustment factor on the weights of respondents who aged 65+ is large and in a downwards direction (the adjusted weights are smaller), as these are the respondents in poorest health. Meanwhile, the adjustment factor does not change the weights of respondents aged between 16 and 64 considerably.

	All respondents a	ged 16+	Respondents ageo	l 16 to 64	Respondents aged 65+		
SHS	Using unadjusted	Using	Using unadjusted	Using	Using	Using	
	weights	adjusted	weights	adjusted	unadjusted	adjusted	
		weights		weights	weights	weights	
Excellent	29.37%	30.28%	29.60%	30.39%	29.42%	30.37%	
	(28.16%-30.58%)		(28.19%-31.01%)		(27.05%-31.79%)		
Good	45.58%	45.95%	46.08%	46.12%	48.02%	49.57%	
	(44.26%-46.90%)		(44.54%-47.62%)		(45.42%-50.62%)		
Fair	16.67%	16.48%	15.60%	15.32%	18.09%	18.07%	
	(15.68%-17.66%)		(14.48%-16.72%)		(16.02%-20.16%)		
Poor	6.47%	6.17%	6.93%	6.65%	2.22%	1.40%*	
	(5.82%-7.12%)		(6.14%-7.72%)		(1.45%-2.99%)		
V. Poor	1.91%	1.12%*	1.79%	1.52%	2.25%	0.59%*	
	(1.55%-2.27%)		(1.38%-2.20%)		(1.48%-3.02%)		
Note: The n	umbers in brackets are 9	5% confidence	e intervals. * indicates that	t the adjusted p	roportion falls out of th	ne CI	

Table 6 Weighted proportions across the categories of subjective health status.

of its corresponding unadjusted proportion (i.e. there is a significant difference between the adjusted and unadjusted proportions).

12.3 Multivariate analysis

The multivariate analysis was carried out in order to investigate factors affecting SHS. This was done by estimating three panel data models (using the whole sample, respondents aged 16 to 64 and respondents aged 65+). To detect the potential effect of the adjustment factor on estimates, each model was estimated two times by using unadjusted and adjusted weights. The comparison between models using unadjusted weights and models using adjusted weights revealed how adjusting the weights would affect the estimation.

In this analysis, the five categories of SHS (excellent, good, fair, poor and very poor) were reorganised. The first three categories were combined into one category (good health status) and the last two were combined into another category (poor health status). Accordingly, SHS became a categorical variable with two categories, indicating whether the respondent has good or poor health status. This variable was used as the dependent variable in the analysis.

$$SHS_i = \begin{cases} 1, & \text{if case i has a good health status.} \\ 0, & \text{if case i has a poor health status.} \end{cases}$$
(16)

Where

 $SHS_i \equiv$ Subjective health status.

The independent variables are sex, ethnicity, age, financial situation, income, marital status, type of household, energy compared to average at the same age, smoking status and number of visits to GP since last year (NVGP). These variables are known for their effect on health status and were used in prior research of self-assessed health in the BHPS (for example Jones *et al*, 2004).

Using data from 18 waves of the BHPS, allowed the estimation of a random effects logistic regression model. However, as described earlier, the model was estimated separately for the whole sample, respondents aged between 16 and 64 and respondents aged 65+. Moreover, the model for each group was estimated with adjusted and unadjusted weights separately.

The analysis was done at the individual level in STATA. The data was introduced as a panel data set to allow the consideration of multiple observations per person and therefore the application of panel data modeling. BHPS sample is not a simple random sample; it was rather selected through a complex sampling design that involved clustering and stratification. However, STATA –like many statistics software– does not support the identification of clustering and stratification while estimating a panel data model. Thus, it was not possible to take clustering and stratification into account in this analysis. This may lead to under estimating the standard errors of estimates. However, although this may not lead to precise standard errors, any differences between the estimates produced using adjusted and unadjusted weights will be due to adjusting the weights since the modeling strategy is held constant. Furthermore, significance levels of regression coefficients were interpreted conservatively and only highly significant coefficients were considered.

Table 7 shows the results of the logistic regression models. The table presents odds ratios. Although the models capture significant relationships between subjective health status and most of the factors included in the analysis, the importance of these factors differs between those aged between 16 and 64 and those aged 65+. For example, in the time period 2003-2008, individuals indicate worse health status than in 1991-1996. However, this is only significant for respondents aged 65+ ($\hat{b}_{16-64,1}$ = 0.714, p > 0.05 and $\hat{b}_{16-64,2}$ = 0.689, p > 0.05; $\hat{b}_{65+,1}$ = 0.561, p < 0.01 and $\hat{b}_{65+,2}$ = 0.523, p < 0.01). However, for the purpose of this paper, the focus is rather on the comparison between the results from unadjusted and adjusted weights. Turning to this, the result of the comparison can be summarised in what follows:

First, focussing on models for respondents aged between 16 and 64, no differences are found between estimates in the model using unadjusted and the model using adjusted weights. In other words, adjusting the weights for respondents aged 16 to 64 does not affect estimates. This result is expected since the adjustment factors in all the age groups under 65 have values close to 1; hence they do not change the weighting much.

Second, as regard to the models of the whole sample and those aged 65+, the significance levels of four estimates were increased when adjusted weights were used to estimate the model. For the whole sample these estimates are related to the following factors:

- *'financially struggling'* $(\hat{b}_1 = 0.477, p < 0.05; \hat{b}_2 = 0.512, p < 0.01).$
- 'single' ($\hat{b}_1 = 0.498$, p < 0.05; $\hat{b}_2 = 0.476$, p < 0.01).
- 'has less energy as average at their age' $(\hat{b}_1 = 0.901, p < 0.05; \hat{b}_2 = 0.889, p < 0.01).$
- 'single-person household' ($\hat{b}_1 = 0.475$, p < 0.05; $\hat{b}_2 = 0.433$, p < 0.01).

As for those aged 65+ the estimates are related to:

- 'age' ($\hat{b}_1 = 0.610, p < 0.05; \hat{b}_2 = 0.587, p < 0.01$).
- *'financially struggling'* $(\hat{b}_1 = 0.578, p < 0.05; \hat{b}_2 = 0.522, p < 0.01)$.
- 'single' ($\hat{b}_1 = 0.651$, p < 0.05; $\hat{b}_2 = 0.628$, p < 0.01).
- 'single-person household' ($\hat{b}_1 = 0.457$, p < 0.05; $\hat{b}_2 = 0.433$, p < 0.01).

As expected, these results suggest no changes in the significance level in the model of those aged 16-64. Additionally, they show that the adjustment factor affects the weights of those aged 65+ mostly and as a result it changes the weights for the sample as a whole. Adjusting the weights results in reducing the values of large weight and therefore minimising the weights' variance. Consequently, the adjusted weights (with less variability) reduce the standard errors of some estimates and hence they become more significant.

Regarding the bias, this was assessed by checking if the coefficients estimated through adjusted weights, falls out of the 95% CIs of the equivalent coefficients in models with unadjusted weights. For ease of exposition, these 95% CIs are not displayed in table 2.7. However, across the three sets of respondents, all of the coefficients in the adjusted models fall within 95% CI of their corresponding coefficients in the equivalent unadjusted models. This means that we have no evidence that our adjustment have reduced the potential bias from the estimated coefficients in the models.

To sum up, in general, estimates constructed based on unadjusted weights are similar to estimates based on adjusted weights. However, some of the estimates increased in their significance level when constructed using adjusted weights. This is particularly in estimates related to respondents aged 65+ as the likelihood of survival (represented by the adjustment factor) in this age group is lower than in their counterparts ages (16 to 64). Corollary, estimates constructed using the whole sample, may also be affected if constructed using adjusted weights. In this analysis, some of the estimates produced based on the whole sample, appeared more significant when constructed using adjusted weights.

As for bias, the multivariate analysis does not provide clear evidence that adjusted weights result in bias reduction. Thus, it cannot be asserted that the adjustment has resulted in bias reduction in our regression coefficients.

Finally, based on this analysis, it can be concluded that most of the ineligible (dead) respondents amongst respondents of unknown eligibility in the BHPS sample are within the age group 65+ rather than within the ages 16 to 64. Thus, if the weighting is not controlled for those aged 65+, the estimated standard errors of some estimates might be misleading.

Table 7 Random	effects 1	ogistic r	egression	models for	or the	determinants	of subj	ective he	ealth status.

	All respondents aged 16+		Aged betwe	en 16 and 64	Aged 65+	
	Using	Using adjusted	Using	Using adjusted	Using	Using adjusted
	unadjusted	weights	unadjusted	weights	unadjusted	weights
	weights		weights		weights	
Year 1997 to 2002	0.862	0.791	0.804	0.812	0.732	0.758
Year 2003 to 2008	0.903	0.921	0.714	0.689	0.561***	0.523***
Female	1.079	1.038	0.788	0.721	1.098**	1.087**
White	1.121	1.047	1.098	1.126	1.079**	1.068**
NVGP	0.735***	0.699***	0.813***	0.791***	0.939***	0.951***
Age	0.610**	0.587***	0.690**	0.637**	0.578***	0.529***
Financially okay	1.301	1.241	1.260	1.199	0.496	0.563
Financially Struggling	0.578**	0.522***	0.715**	0.701**	0.477**	0.512***
Annual income/1000	1.014**	1.047**	1.094**	1.068**	1.030***	1.011***
Single	0.651**	0.628***	0.658**	0.620**	0.498**	0.476***
Widow	0.459	0.486	0.708	0.734	0.405**	0.417**
Divorced or separated	0.794**	0.768**	0.532**	0.510**	1.031**	1.020**
Smoker	0.886***	0.852***	0.762***	0.723***	0.861***	0.823***
Has GCE qualification or above	1.092	1.076	1.013	1.116	0.724	0.715
Has same energy as average at their age	1.187	1.206	1.203	1.192	0.588**	0.539**
Has less energy as average at their age	0.784***	0.723***	0.885***	0.817***	0.901**	0.889***
Single-person household	0.457**	0.433***	0.340**	0.311**	0.475**	0.433***
N rho	5019 0.41	5019 0.43	3758 0.36	3758 0.36	1261 0.30	1261 0.33

Note: Data are from BHPS 1991-2008. Entries are odd ratios. Across the three sets of respondents, all of the odd ratios in the adjusted models fall within 95% CI of their corresponding odd ratios in the equivalent unadjusted model, indicating no significant difference between the coefficients in the two models. The reference categories of the categorical independent variables are: year 1991 to 1996, male, non-white, having good financial situation, married or living with a partner, not a smoker, does not have a GCE qualification or above, has more energy compared to average at their age and multiperson hh respectively. rho represents the percentage of variance that is due to differences across respondents, and the values in the table indicate enough variability between respondents to favour a random effects model. ** p < 0.05, *** p < 0.01.

13. Conclusion

This paper investigates a limitation in the standard methodology of weighting for nonresponse in longitudinal surveys in relation to dealing with units of unknown eligibility whilst weighting (during weighting, the standard approach assumes that all cases for which eligibility is unknown are eligible). The investigation here is based on considering the definition of eligibility in the BHPS (and in most longitudinal social science surveys today) which is 'sample member is alive and living in the country where the survey is conducted'. Using population information on survival/mortality, the alternative method in this paper estimates likelihood of survival (adjustment factor) for cases in the sample (by comparing survival rates in the population with survival rates in the sample). The adjustment factor was then used to adjust non-response weights to correct for the potential inclusion of ineligibles in weights calculation. The findings in this paper suggest the following:

While weighting for non-response endeavours to reduce bias by correcting for the missing units and adjust the distribution of the responding sample to resemble the distribution of the selected sample, unknown eligibility may mislead this process. If some of the unknown eligibility units, that are assumed to be eligible during weighting, are actually ineligible, then weighting, as in the standard approach, may mislead the calculation of survey estimates. It will allow the eligible responding units, that are similar to the ineligible ones, to contribute to the calculation of the estimate in question more than they should. This is especially the case if the cause of ineligibility (e.g very poor health may cause death) is directly linked to what is being measured (e.g. indicators of good health). The findings in this paper appear to support this.

The introduced adjustment factor in this paper reduces the value of some of the weights from the standard approach which were otherwise too large, due to including some ineligible units in the denominator of the weighting model. Consequently, the variance of the weights is also reduced. As a result, the adjusted weights have significantly different impact on some estimates (by reducing their standard errors).

With respect to bias, although the multivariate analysis here does not show evidence that the adjusted weights reduce potential bias, the descriptive findings demonstrate that adjusted weights could result in different estimates of some descriptive statistics. This indicates that adjusted weights may remove bias components that were included because the standard weights contain influence of some ineligible units. If this approach applied in different data/analyses, with the availability of more accurate population information on survival/mortality, evidence for bias reduction are likely to be clearer both on estimates from multivariate and descriptive analyses. Thus, surveys that suffer from high rates of unknown eligibility, and where eligibility is also defined by being alive and living in the geographical area covered by the survey, the method is highly recommended.

However, when this approach is used one should pay extra attention to the mortality rates that are used to calculate the adjustment factor. For accurate calculation of adjustment factor, mortality rates should be up-to-date and reliable. For instance, the availability of a single-year mortality rates from the population (rather than 10-year age band as in the analysis in this paper) would improve the calculation of the adjustment factor as large age bands might conceal some information. Also, more importantly, one should use mortality rates for the same population covered by the survey both in terms of time period and geographical area. For example, in this research, the BHPS sample was selected only from residential addresses, meanwhile registered population mortality statistics include people at all types of addresses (e.g. nursing homes). Thus, registered mortality rates may not perfectly match the rates in the population of interest, at least for the first two or three years of the survey (eventually, those initially institutionalised people will die, and all of the new institutionalised population will have been from the residential addresses covered by the survey, so at that point the survey should become representative of the entire population, i.e. the same population to which the mortality statistics refer).

Moreover, the availability of population information on emigration and institutionalised individuals would be beneficial. Combining this information with mortality rates when calculating the adjustment factor will result in more accurate values as all forms of ineligibility are taken into account.

However, in surveys where ineligibility predominantly occurs by satisfying other characteristics (e.g. reaching a specific age or belonging to a certain social group) and maybe partially through death, the strategy of survival/death based adjustment factor may not be very useful. This is because the calculation of the adjustment factor (which is based on comparing the survival proportions in the sample and the population) in this case will have not taken into account the main forms of ineligibility. The approach of the adjustment factor will be more effective if the main ineligibility form in the sample can be found in the records of population statistics, or other reliable external data, as the case in this paper (i.e. population mortality rates).

Finally, an alternative procedure is to carry out a case-level death imputation in trying to identify dead sample members. This can be achieved by applying some kind of imputation model that can use available information about respondents of unknown eligibility (from their last interview), and relate it to similar characteristics of dead respondents. This can assist in identifying individuals among unknown eligibility cases who possess the same characteristics as dead respondents. For example, indicators of old age and poor health may be strong candidate variables for the imputation model. On the one hand, in this approach, eligibility is identified based on information from within the survey (no need for seeking and utilising information from external source such as a census etc...). Moreover, the main advantage of this method is that it helps in determining eligibility at the case level. Thus, weights' construction will easily exclude ineligible units. However, this approach may not be accurate, since death can sometimes occur randomly (i.e. regardless of age and health status). Therefore, the imputed values would contain some random variation as the imputation model is imperfect. Nonetheless, in situations where information on mortality rates in the population is not available/reliable, this approach could be recommended.

Another alternative procedure (used in the Health Survey for England and HILDA) could be contacting the death register office. In almost every country there is an office where deaths are registered. These offices collect information such as name, time and date of death, place and date of birth, the last address, occupation, reason for death and contact information of a surviving person related to the person died (usually a spouse or civil partner). If the survey organisation is able to contact the death register office and obtain this information, death can be identified by matching the records of respondents of unknown eligibility with the information held in the register office. The advantage of this approach is that it produces precise estimates based on accurate information. However, apart from the fact that this approach is time consuming (need to be done for all unknown eligibility cases at every wave), in some countries, register offices may not be willing to co-operate, for reasons of confidentiality. In any case, for the BHPS sample, any method of identifying death in the sample should focus on respondents aged 65+ as most of the unknown death is centred in this age group.

References

AAPOR (American Association for Public Opinion Research). (2011) "Standard Definitions: final dispositions of case codes and outcome rates for surveys". 7th edition. AAPOR.

Barron, M., Khare, M., and Zhao, Z. (2008) "Cell Telephone Response Rates", Section on Survey Research Methods- 2008 AAPOR. [online] available from <u>http://www.amstat.org/Sections/Srms/Proceedings/y2008/Files/barronmartin.pdf</u> [Accessed 18th May 2011].

Becketti, S., Gould, W., Lillard, L., and Welch, F. (1988) "The Panel Study of Income Dynamics after Fourteen Years: an evaluation", *Journal of Labour Economics*, vol 6, pp. 472-492.

Biemer, P., and S. L. Christ. (2008) "Weighting Survey Data", (chapter 17) in De Leeuw, E.D., J.J. Hox and Dillman, D.A., *International Handbook of Survey Methodology*. New York: Lawrence Erlbaum. **Chang**, M. (2010) "Adjusting for Nonresponse Bias Due to Attrition and Intermittent Wave Nonresponse in Longitudinal Surveys of Older Adults", *Dissertation*. Nebraska: University of Nebraska.

Deane, G. (1990) "Mobility and Adjustments: paths to the resolution of residential stress", *Demography*, vol.27, pp. 65–79.

Dr Foster. (2004) "Death Rates in England and Wales and the United States: variation with age, sex and race", *BMJ*, vol.329, pp. 1367-1367.

Ellis, J. (2000) "Estimating the Number of Eligible Respondents for a Telephone Survey of Low-Incidence Households". [online] available from

http://www.amstat.org/sections/srms/proceedings/papers/2000_180.pdf [Accessed 12th June 2011].

Fitzgerald, J., Gottschalk, P., and Moffitt, R. (1998) "An Analysis of Sample Attrition in Panel Data: the Michigan Panel Study of Income Dynamics", *Journal of Human Resources*, pp. 251-299.

Frankel, R. (1983) "The Report of the CASRO Task Force on Response Rates", (pp. 1-11) in Wiseman, F., *Improving Data Quality on Sample Surveys*. Cambridge: Marketing Science Institute.

Frankel, R., Battaglia, P., Kulp, W., Hoaglin, C., Khare, M., and Cardoni, J. (2003) "The Impact of Ring-No-Answer Telephone Numbers on Response Rates in Random-Digit Dialing

Surveys", Presentation to the American Statistical Association, San Francisco.

Gasteiz, V. (2007) "Standardisation and Systematisation of Response Rate Calculation", *Euskal Estatistika Erakuundea, Basque Statistics Office*. [online] available from <u>http://www.eustat.es/document/datos/ct_16_i.pdf</u> [Accessed 7th May 2011].

Gray, R., Campanelli, P., Deepchand, K., and Prescott-Clarke, P. (1996), "Exploring Survey Non-response: the effect of attrition on follow-up of the 1984-85 health and life style survey", *The Statistician*, vol. 45, pp. 163-183.

<u>Groves, R.</u>, Fowler, F., Couper, <u>M., James M., Lepkowski</u>, J., <u>Singer</u>, E., and Tourangeau, R. (2004) "*Survey Methodology*", New York: Wiley.

Groves, R. (2006) "Nonresponse Rates and Nonresponse Bias in Household Surveys", *Public Opinion Quarterly*, vol. 70, pp. 646-675.

Jones, A., Koolman, X., and Rice, N. (2004) "Health-related Attrition in the BHPS and ECHP: using inverse probability weighted estimators in nonlinear models", [online] available from <u>http://www.ccsr.ac.uk/methods/events/attrition/documents/Jones.pdf</u> [Accessed 17th July 2013].

Kroh, M. (2009) "Data Documentation: sample sizes and panel attrition in the German Socio Economic Panel (1984 until 2008)", Berlin: Deutsches Institut für Wirtschaftsforschung (DIW).

Laurie, H., Smith, R., and Scott, L. (1999) "Strategies for Reducing Nonresponse in a Longitudinal Panel Survey" *Journal of Official Statistics*, vol.15, pp. 269-282.

Laurie, H., and Lynn, P. (2009) "The Use of Respondent Incentives on Longitudinal Surveys", (chapter 12) in Lynn, P., *Methodology of Longitudinal Surveys*. Sussex: John Wiley & Sons, Ltd.

Lepkowski, J., and Couper, M. (2002) "Nonresponse in the Second Wave of Longitudinal Household Surveys", in Groves, R., Dillman, D., Eltinge, J., and Little, R., *Survey Nonresponse*. New York: John Wiley.

Lessler, J., and Kalsbeek, W. (1992) "Nonsampling Error in Surveys", New York: John Wiley & Sons.

Little, R. (1986) "Survey Nonresponse Adjustments for Estimates of Means", *International Statistical Review*, vol. 54, pp.139-157.

Lynn, P. (1996) "Weighting for Non-response", pp. 205-214 in Banks. R. et al. Survey and statistical computing 1996, ASC.

Lynn, P. (2005) "Weighting", pp.967-973 in Kempf-Leonard. K. *Encyclopedia of Social Measurment*, Academic press.

Lynn, P. (2009) "Methods for Longitudinal Surveys", (chapter 1) in Lynn, P., *Methodology of Longitudinal Surveys*. Sussex: John Wiley & Sons, Ltd.

Lynn, P. (2011) "Maintaining Cross-Sectional Representativeness in a Longitudinal General Population Survey", *Institute for Social and Economic Research Working Papers*, No. 2011-04. Colchester: University of Essex.

McGonagle, K., Couper, M., and Schoeni, R. (2011) "Keeping Track of Panel Members: an experimental test of a between-wave contact strategy", *Journal of Official Statistics*, vol.27, No. 2, pp. 319-338.

McHugh, K. E., Gober, P., and Reid, N. (1990) "Determinants of Short- and Long- Term Mobility Expectations for Home Owners and Renters", *Demography*, vol.27, No. 1, pp. 81-95.

Nicoletti, C., and Buck, N. (2004) "Explaining Interviewee Contact and Co-operation in the British and German Household Panels", (pp. 143-166) in Ehling, M., and U. Rendtel *Harmonisation of Panel Surveys and Data Quality*. Wiesbaden: Statistisches Bundesamt.

Nicoletti, C., and Peracchi, F. (2005) "Survey Response and Survey Characteristics: microlevel evidence from the European Community Household Panel", *Journal of the Royal Statistical Society*, Series A, vol. 168, pp. 763-781.

ONS. (2014) "Deaths Registered in England and Wales, 2013", *Statistical Bulletin*, [online] available from http://www.ons.gov.uk/ons/dcp171778_370815.pdf .

Plaza, S., and E., Graf. (2008) "Executive Summary: recommendations and practical examples for using weighting", Lausanne: University of Lausanne.

Rabe, B., and Taylor, M. (2009) "Residential Mobility, Neighbourhood quality and lifecourse events", *Institute for Social and Economic Research Working Papers*, No. 2009-28. Colchester: University of Essex.

Sadig, H. (2011) "Non-response Weight Adjustments in Longitudinal Surveys" (pp.207-210) in *Survey Research Methods and* Applications, Pisa: Edizioni plus-Pisa University Press.

Singh-Manoux, A., Gueguen, A., Ferrie, J., Shipley, M., Martikainen, P., Bonenfant, S., Goldberg, M., and Marmot, M. (2008) "Gender Differences in the Association Between

Morbidity and Mortality Among Middle-Aged Men and Women", *American Journal of Public Health*, vol. 98, No. 12, pp. 2251-2257.

Smith, T. (2003) "Areview of Methods to Estimate the Status of Cases with Unknown Eligibility", *Report Presented to AAPOR, Phoenix, May, 2004.*

Speare, A. (1974) "Residential Satisfaction as an Intervening Variable in Residential Mobility", *Demography*, vol.11, No. 2, pp. 173-188.

Survey Research Centre (SRC). (2005) "Notes on Response Rate", Waterloo: University of Waterloo. [online] available from

http://www.src.uwaterloo.ca/Services/SRCResponseRates.pdf [Accessed 4th May 2011].

Taylor, M. (ed.). (2010) "British Household Panel Survey User Manual Volume A", Colchester: Institute for Social and Economic Research, University of Essex.

Uhrig, N. (2008) "The Nature and Causes of Attrition in the British Household Panel Survey", *Institute for Social and Economic Research Working Papers*, No. 2008-05. Colchester: University of Essex.

Watson, D. (2003) "Sample Attrition between Waves 1 and 5 in the European Community Household Panel", *European Sociological Review*, vol. 19:4, pp.361-378.